

# Temi di discussione

(Working Papers)

The IRB approach and bank lending to firms

by Raffaele Gallo

bag 1347



# Temi di discussione

(Working Papers)

The IRB approach and bank lending to firms

by Raffaele Gallo

Number 1347 - October 2021

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Luca Citino, Francesco Franceschi, Danilo Liberati, Salvatore Lo Bello, Juho Taneli Makinen, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

# THE IRB APPROACH AND BANK LENDING TO FIRMS

# by Raffaele Gallo<sup>\*</sup>

#### Abstract

This paper examines the impact of the regulatory approach adopted to calculate capital requirements on banks' lending policies. Since the capital absorption of loans to high-risk borrowers is greater under the internal ratings-based (IRB) method than under the standardized approach (SA), IRB banks may raise interest rates and reduce credit to riskier borrowers following their regulatory regime shift. The analysis examines banks' lending policies around each of the shifts that occurred between 2007 and 2017. First, in a context of declining rates and credit growth, banks adopting the IRB approach decreased interest rates (credit) less (more) for riskier than for safer borrowers when compared with SA intermediaries. Second, an existing credit relationship with a high-risk borrower is more likely to end after the shift. Third, the results at the firm level suggest that high-risk borrowers partly compensated the reduction in bank credit by obtaining funds from SA institutions, but that they were not able to offset the rise in their average cost of credit because of the significant costs involved in switching lenders.

## JEL Classification: G20, G21, G28, G32.

**Keywords**: credit risk regulation, interest rates, bank credit, internal rating model. **DOI**: 10.32057/0.TD.2021.1347

#### Contents

1. Introduction	5
2. Review of regulatory background and related research	8
3. The research hypotheses	9
4. Data and sample	10
5. Methodology and descriptive statistics	12
5.1 Methodology	12
5.2 Descriptive statistics	14
5.3 Preliminary evidence: share of high-risk loans	15
6. Results	15
6.1 The impact of the regulatory regime shift on the cost and the amount of credit	15
6.2 The impact of the regulatory regime shift on the extensive margin of credit	16
6.3 The impact of the regulatory regime shift at the firm level	17
7. Additional tests	18
7.1. Borrowers' characteristics	18
7.2. Falsification test	19
8. Conclusions	19
Appendix	21
References	22
Figures and tables	24

<sup>\*</sup> Bank of Italy, Directorate General for Economics, Statistics and Research.

## **1. Introduction**<sup>1</sup>

The entry into force of Basel II in 2007 has significantly changed the assessment of credit risk. In this framework banks are allowed to choose between the standardized approach (SA), which assigns to each exposure a fixed risk weight, and the internal ratings-based (IRB) approach, which makes capital requirements a function of banks' internal estimates about the riskiness of the borrower.

The adoption of the IRB approach has several and widespread effects on banks' risk management and activities. In general, internal risk models are associated with a high complexity and a certain discretion that make risk estimates comparison difficult across financial institutions (Cannata et al., 2012; Cucinelli et al., 2018; Le Leslé and Avramova, 2012; Mariathasan and Merrouche, 2014; Pérez Montes et al., 2018). These implications are at the core of supervisory scrutiny and the international debate, especially in view of the future implementation of the next regulatory framework (BCBS, 2017).

This work focuses on the impact of these approaches on the sensitivity of bank capital to asset risk: since capital charges are more linked to asset risk when intermediaries adopt internal models, the regulatory capital is more risk-sensitive for IRB banks than for SA ones (Behn et al., 2016b; Kashyap and Stein, 2004; Repullo and Suarez, 2004, 2013).

The IRB approach provides a capital incentive for banks with good risk management. Indeed, this framework would require less capital for less-risky assets than the standardized approach and more capital for riskier assets (Le Leslé and Avramova, 2012). In particular, the two approaches lead to significant differences in the treatment of exposures to unrated firms (Repullo and Suarez, 2004). Under SA, all exposures to unrated borrowers are generally associated with a fixed risk weight (i.e. 100 per cent); under IRB, the risk weights are lower than 100 per cent for safer borrowers, while are higher for riskier ones. Consequently, the capital absorption of loans to low-risk (high-risk) borrowers is lower (greater) for IRB than for SA banks. This asymmetric treatment affects the cost of each exposure for banks (i.e. its capital absorption), and may have a significant impact on their pricing and credit policies. After the adoption of the IRB approach, banks may pass-through the increase in cost by raising interest rates to riskier borrowers. They also have an incentive to reallocate credit from riskier to safer assets to exploit savings in capital charges. Overall, banks adopting the IRB approach may raise interest rates and reduce credit to high-risk borrowers with respect to low-risk ones.

The main goal of this paper is the assessment of the impact of the transition from the standardized approach to the IRB method on Italian banks' lending policies. The sample includes unrated

<sup>&</sup>lt;sup>1</sup> I thank G. Cascarino, F. Ciocchetta, S. De Mitri, R. Felici, and M. Moscatelli for their support in data collection. The work benefited from the useful comments of L. Carpinelli, F. Columba, A. De Vincenzo, D. Drago, G. Guazzarotti, S. Magri, and F. Palazzo.

borrowers with at least one loan granted by a "treated" bank (i.e. those adopting the IRB approach in  $t_0$ ) and a "control" bank (i.e. those that have not changed their approach in  $t_0$ ) between 2007 and 2017. The sample of treated banks consists of the 8 Italian institutions that have adopted the IRB approach in the analyzed period. The share of total bank lending to firms granted by Italian IRB intermediaries reached 65 per cent in 2017. This statistic suggests that, since the regulatory regime shift is associated with substantial administrative and organizational costs, mainly larger banks have started to use internal rating models; nevertheless, a significant share of credit in Italy is still extended by SA banks.

The main identification strategy relies on the examination of borrowers that receive credit immediately before and after the regulatory regime switch of their lender. The analysis focuses on the supply of credit by examining the same firm that receive credit from banks adopting different regulatory approaches in the same period. This setup allows saturating specifications with firm-time and bank-time fixed effects to absorb all time-varying (observed and unobserved) firm and bank heterogeneity (Jiménez et al., 2014). By accounting for loan demand (consistent with Khwaja and Mian, 2008) and time-varying bank heterogeneity, identification resides in comparing changes in the cost and the amount of credit around each bank-specific regulatory regime shift date by the same bank to firms that differ in their riskiness level (measured by relying on an accounting-based credit score).

First, this framework allows verifying the impact of the adoption of the IRB approach on the cost and the amount of bank credit by comparing safer and riskier firms. Second, the analysis focuses on the "extensive margin" of credit, defined as the number of borrowers before and after the shift, by examining whether existing relationships with IRB banks are more likely to end if the loan is granted to a riskier firm. Finally, the paper examines the effect of banks' regulatory approach changes on the ability of riskier firms to access funds, verifying at the firm level whether high-risk borrowers are able to offset the potential negative effects on the cost and the amount of credit associated with the shift of their lender by relying on SA banks.

The results show that, in a period characterized by an average decline in the cost of credit and the credit growth for all borrowers, treated banks decreased interest rates (credit) less (more) for riskier than for safer borrowers when compared with SA intermediaries. The additional effect of the shift on the change in the cost of credit for riskier borrowers is 4 basis points, while the reduction in the credit growth is 1 percentage point larger. For high-risk firms, existing credit relationships are also more likely to end after the regulatory shift of their lenders.

Overall, the analysis at the firm level suggests that riskier firms partly compensated the reduction in bank credit by obtaining funds from SA banks, but they were not able to avoid a relative increase in their average cost of credit. Indeed, riskier borrowers of a treated bank experienced a decrease in the average interest rates that is 43 bps lower than that faced by safer companies; this effect is 17 bps higher than that observed for borrowers of SA banks in the same period (26 bps).

This work contributes to the literature on the impact of the IRB approach on bank lending (Behn et al., 2016b; Repullo and Suarez, 2004, 2013). To the best of my knowledge, this paper is the first to analyze empirically how the regulatory approach adopted by the bank affects the cost and the amount of credit of their borrowers around the shift.

In particular, this work is closely related to the theoretical contribution of Repullo and Suarez (2004). The empirical evidence presented in this work is in line with their prediction on the reduction in the cost of credit for low-risk borrowers of banks adopting the IRB approach. In contrast, the results related to the relative increase in rates for high-risk borrowers are not consistent with their expectations, which indicate that, in a perfectly competitive market for business loans, riskier firms may avoid increases in their rates by borrowing from lenders that adopt the less risk-sensitive SA. This difference may be due to the costs for borrowers involved in switching to different lenders (Rajan, 1992; Sharpe, 1990), which may be particularly significant for riskier firms. Indeed, the results obtained at the firm level suggest that riskier borrowers that switched from an IRB to a SA bank after the shift have obtained credit at higher rates from the latter lender, consistent with the presence of significant switching costs.

Other related works focus on how banks adopting different regulatory approaches react to an exogenous shock on the demand side, e.g. an economic downturn or a worsening of financial market conditions (Behn et al., 2016b; Bruno et al., 2017; Cucinelli et al., 2018; Gallo, 2020). These studies do not explore the effect on bank lending directly associated with the regulatory regime switch (i.e. they do not compare lending policies of the same bank before and after its regulatory shift). Moreover, the main analysis in this paper is at the bank-firm level to exploit the information advantage of granular data, while other studies focus on more aggregate measures, such as banks' risk-weighted assets (Bruno et al., 2017) or the NPL ratio (Cucinelli et al., 2018) as a proxy of the level of risk borne by a bank.

The remainder of this paper is organized as follows. Section 2 reviews the related literature and outlines the regulatory background. Section 3 presents the research hypotheses. Section 4 describes the dataset. Section 5 discusses the methodology and presents a preliminary evidence relying on data aggregated at the bank level. The main results are reported in Section 6. Section 7 presents additional tests. Section 8 concludes.

# 2. Review of regulatory background and related research

After the introduction of Basel II, the required bank capital is calculated as a fraction of riskweighted assets (RWA), which are estimated by multiplying the value of each bank asset by a corresponding risk weight. Under the standardized approach, fixed risk weights are defined for each bucket of assets, similarly to the Basel I framework. Under this method, uncollateralized loans granted to unrated firms have a risk weight of 100 per cent.<sup>2</sup> Therefore, the assigned risk weight is not sensitive to the riskiness of the borrower.

In contrast, the risk weight of each asset under the IRB approach depends on banks' internal risk models. Internal estimates rely on four parameters: the probability of default (PD), the loss given default (LGD), the exposure at default (EAD), and the effective maturity of the loan.<sup>3</sup> Each model has to be approved and validated at least once a year by regulators. The estimates under IRB have to be periodically updated. Consequently, risk weights and related capital charges for each exposure may change over time.<sup>4</sup>

The literature has discussed the incentives stemming from the adoption of the IRB approach. First, a high responsiveness of capital requirements to a change in the borrower's riskiness may create incentives for banks to deleverage after a shock, amplifying business cycle fluctuations (Behn et al., 2016b; Kashyap and Stein, 2004; Repullo and Suarez, 2013) and financial instability (EBA, 2013). On the other hand, the reduction in the loan portfolio riskiness after a deterioration of credit conditions may strengthen the solvency of the banking system in the long-term (Repullo and Suarez, 2013).

In addition, the adoption of internal models allows banks to exert greater discretion in the assessment of the borrower's riskiness. Some evidence suggests that the adoption of the IRB approach may provide intermediaries some incentives to underestimate the riskiness of their assets to reduce capital requirements (Behn et al., 2016a; Plosser and Santos, 2014).

The literature on the impact of the IRB approach on credit supply shows that IRB banks adopt different lending policies than SA intermediaries. After a shock to the average riskiness of borrowers (e.g. economic downturns), IRB banks reallocated credit from riskier to safer borrowers more than SA intermediaries. Behn et al. (2016b) find that the amount of IRB loans granted by German banks decreased more than that of SA loans in response to an exogenous shock to credit risk (i.e. the collapse

 $<sup>^{2}</sup>$  In contrast, the risk weight assigned to assets rated by a specialized agency can change after a downgrade or an upgrade of the external credit rating. Moreover, the risk weight may be lower than 100 per cent for specific exposures, such as SMEs.

<sup>&</sup>lt;sup>3</sup> Banks estimate all four parameters under the advanced IRB (AIRB) approach, while they estimate only the PD in the foundation IRB (FIRB) approach and assume fixed standard values for the other parameters. This distinction does not affect the main findings because the treatment for riskier and safer borrowers is equal under both methods.

<sup>&</sup>lt;sup>4</sup> The differences in capital absorption between IRB and SA for each bucket of assets are significant but, at the bank level, they can be partly smoothed by Pillar 2 requirements.

of Lehman Brothers in September 2008). Bruno et al. (2017) find that IRB banks have reduced more their corporate loan portfolio than SA ones between 2008 and 2014, both in absolute terms and as a share of total credit exposures. Cucinelli et al. (2018) show that, in the aftermath of the financial and economic crisis (2008-2015), European IRB banks adopted more conservative lending policies by granting new loans to safer customers. Finally, Gallo (2020) finds that the adoption of the IRB approach made loan rates applied to listed firms significantly more market-oriented: a worsening of financial market conditions is associated with a rise in interest rates applied to listed borrowers greater on loans granted by IRB banks than on those extended by SA intermediaries.

### 3. The research hypotheses

Since the capital absorption of loans to high-risk borrowers is greater for IRB banks than for SA ones, banks adopting internal rating models may pass-through the rise in cost to riskier borrowers and may reallocate credit from riskier to safer assets to save capital after their shift. This main prediction leads to three different testable hypotheses.

First, a shift in the regulatory approach adopted by the lender may affect the cost and the amount of bank credit for safer and riskier firms differently. Holding all else equal, banks adopting the IRB approach may apply higher interest rates and grant less credit to high-risk firms. Therefore, the first testable hypothesis is:

# **H1.** After the adoption of the IRB method, banks raise interest rates and reduce credit more for high-risk firms than for low-risk ones.

Second, a shift in the regulatory approach may have an impact on the probability of ending for existing credit relationships. Indeed the impact of the regulatory regime shift may not be limited to the "intensive margin" of credit (the amount granted to the same borrower before and after the shift) but it also affects the "extensive margin" (the number of borrowers before and after the shift). Consequently, the second hypothesis is that:

**H2.** After the adoption of the IRB method, an existing credit relationship is more likely to end if the loan is granted to a riskier firm with respect to a safer one.

Finally, the third hypothesis focuses on the overall impact on bank credit of the IRB approach adoption. Indeed, a riskier borrower may be able to offset the increase in the cost and the reduction in the amount of credit assumed in H1 and H2 by relying on other banks that have not changed their regulatory approach. If the borrower obtains more credit at a lower interest rate from other banks, the overall impact of the regulatory shift would be negligible at the firm level.

However, switching lenders may lead to substantial costs for borrowers. Indeed, prolonged credit relationships mitigate the significant information asymmetries in the credit market by allowing banks

to accumulate information about borrowers' creditworthiness over time (Boot, 2000). Since this knowledge is bank-specific, firms that want to change their lenders may incur substantial switching costs, which may significantly decrease the access to alternative source of funds. These costs may be higher for more opaque borrowers (Rajan, 1992; Sharpe, 1990) such as riskier firms. As a result, after the regulatory shift of the lender, riskier borrowers might not be able to mitigate the increase in the cost of funding as well as the reduction in the total amount of credit by relying on other banks. Given these premises, the third hypothesis is that:

**H3.** *Riskier borrowers of a lender adopting the IRB method experience an increase in the average cost of credit and a reduction in the total bank credit growth after the regulatory regime shift.* 

#### 4. Data and sample

The sample consists of quarterly data on credit to Italian non-financial firms from the Italian Credit Register ("Centrale dei Rischi", CR) from 2007 to 2017. This database is handled by the Bank of Italy and covers the population of individual borrowers' outstanding exposure of over  $\notin$ 30,000 with a single intermediary.<sup>5</sup> CR provides information on lenders and borrowers, while data on interest rates are obtained from a section of the CR, TAXIA. This latter database contains information on the interest rate charged on all loans granted by a representative sample of Italian banks, which accounts for more than 80 per cent of total bank lending in Italy.<sup>6</sup>

Credit is aggregated at the banking group level because lending policies are typically decided at this level. To ensure the comparability of results, the empirical analysis focuses on the amount of granted credit lines (i.e. the sum of outstanding and loan commitments) at the end of the fourth quarter of each year. This type of credit is the most comparable because it is highly standardized across banks (Sette and Gobbi, 2015), has no specified maturity, is granted for no specific purpose, and is typically not backed by collateral. In addition, since the usage of available credit lines is strongly seasonal (Jiménez et al., 2009; Sette and Gobbi, 2015), the analysis compares the same period of subsequent years.

The change in the net interest rate on revolving credit lines and the growth rate of the granted amount are retrieved for each bank-firm relationship. The interest rate for each credit line is estimated by dividing the amount due by the product between the loan amount and the days this amount was outstanding. To avoid outliers affecting the results, the change in the interest rate and the credit growth are winsorized at the 5th and 95th percentiles and at the 1st and 99th percentiles, respectively.<sup>7</sup>

<sup>&</sup>lt;sup>5</sup> For CR, the reporting threshold was €75,000 before 2009.

<sup>&</sup>lt;sup>6</sup> For TAXIA, the reporting threshold is €75,000.

<sup>&</sup>lt;sup>7</sup> Consistent with Sette and Gobbi (2015), an higher threshold is adopted for the cost of credit to address the greater risk of outliers deriving from the construction of this variable.

Firm accounting data are taken from the database of Cerved group, which is a leading provider in Italy for information on limited companies. For each firm Cerved computes annually a credit score, which is an indicator of the probability of default based on the Z-score methodology (Altman et al., 1994). The Z-score varies from 1 (safest) to 9 (riskiest). Firms with a credit score equal or above six are considered as high-risk borrowers. Data on banks' consolidated balance sheets are obtained from the supervisory reports.

Loans to firms with no balance sheet information in the Cerved database, nonperforming loans, and those extended to borrowers with a rating assigned by a credit rating agency<sup>8</sup> are excluded from the final sample.

The regulatory approach adopted by each Italian bank is singled out by relying on confidential supervisory information and publicly available banks' Pillar 3 reports. Figure 1 shows the number of IRB banks by year in the examined period. The intermediaries that have adopted the IRB approach in year  $t_0$  (8) represent the sample of "treated" banks. The other lenders (553) are included in the sample of "control" banks. Figure 2 presents the share of total bank lending to non-financial firms granted by Italian IRB intermediaries. The share of credit granted by IRB banks ranged from about 40 to 65 per cent over the analyzed period.

The main sample includes firms with at least one loan granted by a treated and a control bank both before and after the shift. For each  $t_0$  (i.e. a year in which at least one bank has adopted the IRB approach), the analysis compares the credit granted to the same borrower in the fourth quarter of  $t_{-1}$  (the year before the shift) and  $t_{+1}$  (the year after the shift). This allows comparing how treated and control lenders evaluate the same borrower, overcoming potential identification issues.

Overall, the sample employed in the following analyses at the bank-firm level includes 599,418 bank-firm relationships (117,348 firms), of which 301,255 are relationships between banks and high-risk borrowers (68,795 firms).

For the exercises at the firm level, the sample consists of companies with at least one loan outstanding both before and after the shift. Therefore, these analyses consider also borrowers with a single credit relationship to assess the overall impact of the regulatory regime shift on the Italian credit market. For these exercises, the sample includes 1,048,032 borrower-time observations (437,820 firms), of which 553,427 are related to high-risk borrowers (296,114 firms).

<sup>&</sup>lt;sup>8</sup> These loans are excluded because the standardized approach prescribes specific risk weights for exposures to rated firms. Therefore, differences in loan contract terms between rated borrowers of SA and IRB banks should be due to a different treatment in the current regulatory framework.

#### 5. Methodology and descriptive statistics

## 5.1 Methodology

To test the impact of the adoption of the IRB approach on the cost and the amount of granted credit to safer and riskier firms (H1), first, the baseline models described in Eqs. (1) and (2) are estimated by employing the sample at the bank-firm level described in Section 4.

$$\Delta Rate_{ijt+1} = \beta_1 HighRisk_{it} + \beta_2 IRBshift_{jt} + \beta_3 HighRisk_{it} \cdot IRBshift_{jt}$$
(1)  
+  $\gamma Controls_{iit} + \theta_{t+1} + \varepsilon_{iit+1}$ 

$$\Delta Credit_{ijt+1} = \beta_1 HighRisk_{it} + \beta_2 IRBshift_{jt} + \beta_3 HighRisk_{it} \cdot IRBshift_{ijt}$$
(2)  
+  $\gamma Controls_{ijt} + \theta_{t+1} + \varepsilon_{ijt+1}$ 

The dependent variables of Eqs. (1) and (2) are  $\Delta Rate$ , which is the difference between the annualized interest rate on granted credit lines in  $t_{+1}$  and  $t_{-1}$  by bank *j* to firm *i*, and  $\Delta Credit$ , which is the difference between the log amount of granted credit lines in  $t_{+1}$  and  $t_{-1}$  by bank *j* to firm *i*, respectively.

*HighRisk* is a dummy variable equal to 1 if the firm's credit score in  $t_0$  is greater than 5 and 0 otherwise, while *IRBshift* is a dummy variable equal to 1 if the bank *j* has adopted the IRB approach in  $t_0$  and 0 otherwise. The variable of interest is the interaction between *HighRisk* and *IRBshift*.

The model also includes a vector of control variables (*Controls*). First, the potential effects of bank-firm relationships are taken into account by introducing *Share*, the share of total credit (revolving credit lines, term loans, and loans backed by accounts receivables) granted by bank *j* to firm *i* in  $t_{+1}$ . A higher value of this variable indicates stronger bank-firm relationships because holding a large share of the overall credit granted to the firm generally allows having better access to significant information (Elsas, 2005).

Second, the borrower's characteristics are controlled by adding firms' accounting data (*Size*, *Leverage*, *Fixed Assets*, *EBITDA*, *IntCovRatio*), *Industry* (indicator variables for the industry of firm *i* based on 2-digit ATECO), as well as *Area* (indicator variables for geographical area of the country).

Third, this vector also includes a set of bank characteristics to take into account lenders' heterogeneity in terms of size (*BankSize*), business model (*Loans* and *Deposits*), capital (*CapRatio*), and profitability (*ROE*). Finally, both models include year fixed effects ( $\theta$ ). Table A.1 in the Appendix presents the complete list of variables with their relative sources.

Afterwards, the equations are estimated by including bank-time ( $\delta$ ) and firm-time ( $\eta$ ) fixed effects, as described in Eqs. (3) and (4). In these equations, fixed effects absorb all time-varying (observed and unobserved) firm and bank heterogeneity. Therefore, both models focus on the differential impact

of a shift in the regulatory approach on changes in the cost and the amount of credit between safer and riskier firms that are contemporaneously borrowers of treated and control banks. Consistent with *H1*, the coefficient of the interaction between *HighRisk* and *IRBshift* should be significant and positive (negative) in the cost (the amount) of credit model.

$$\Delta Rate_{ijt+1} = \beta_1 HighRisk_{it} \cdot IRBshift_{ijt} + \delta_{jt+1} + \eta_{it+1} + \varepsilon_{ijt+1}$$
(3)

$$\Delta Credit_{ijt+1} = \beta_1 HighRisk_{it} \cdot IRBshift_{ijt} + \delta_{jt+1} + \eta_{it+1} + \varepsilon_{ijt+1}$$
(4)

The second analysis focuses on the extensive margin of credit (*H2*). In this case the sample includes firms with at least one loan granted by a treated and a control bank in  $t_{-1}$ . Eqs. (5) and (6) are estimated by adopting *Exit*, which is a dummy variable equal to 1 for loans that existed in  $t_{-1}$  and ceased to exist after the shift, as the dependent variable. A linear probability model is estimated to include bank-time and firm-time fixed effects. Consistent with *H2*, the coefficient of the interaction between *HighRisk* and *IRBshift* should be significant and positive, implying a greater probability that a credit relationship with a treated bank is terminated for riskier firms.

$$Exit_{ijt+1} = \beta_1 HighRisk_{it} + \beta_2 IRBshift_{jt} + \beta_3 HighRisk_{it} \cdot IRBshift_{ijt}$$
(5)  
+  $\gamma Controls_{ijt} + \theta_{t+1} + \varepsilon_{ijt+1}$ 

$$Exit_{ijt+1} = \beta_1 HighRisk_{it} \cdot IRBshift_{ijt} + \delta_{jt+1} + \eta_{it+1} + \varepsilon_{ijt+1}$$
(6)

Both models are estimated by excluding credit relationships that existed in 2007: since numerous relationships were terminated after the global financial crisis, this event may affect the results of the shifts occurred in 2008.<sup>9</sup>

Finally, an analysis at the firm level is employed to verify *H3*. Eqs. (7) and (8) adopt  $\Delta AvgRate$ , which is the difference between the average annualized rate on granted credit lines in  $t_{+1}$  and  $t_{-1}$  to firm *i*, and  $\Delta TotCredit$ , which is the difference between the log amount of total granted credit lines in  $t_{+1}$  and  $t_{-1}$  to firm *i*, as dependent variables, respectively.

$$\Delta AvgRate_{it+1} = \beta_1 HighRisk_{it} + \beta_2 IRBexp_{it} + \beta_3 HighRisk_{it} \cdot IRBexp_{it}$$
(7)  
+  $\gamma Controls_{it} + \theta_{t+1} + \varepsilon_{it+1}$ 

<sup>&</sup>lt;sup>9</sup> Unreported robustness checks show that the main findings are robust to estimating also the other presented models by separating crisis and "normal" periods.

$$\Delta TotCredit_{it+1} = \beta_1 HighRisk_{it} + \beta_2 IRBexp_{it} + \beta_3 HighRisk_{it} \cdot IRBexp_{it}$$

$$+ \gamma Controls_{it} + \theta_{t+1} + \varepsilon_{it+1}$$
(8)

To assess the overall impact of the adoption of the IRB approach, both models adopt a sample that includes also firms with a single credit relationship, as described in Section 4. In both equations the vector *Controls* includes the variable *ShareHHI*, which is calculated as the Herfindahl-Hirschman index (HHI) of share of total credit.<sup>10</sup> The variable of interest is the interaction between *HighRisk* and *IRBexp*, a dummy variable equal to 1 for firms that in  $t_{.1}$  had a share of credit granted by a treated bank falling in the top quartile of the distribution and 0 otherwise.<sup>11</sup> This variable identifies the firms that had a significant credit relationship with a treated bank before its shift, and consequently they should be more "exposed" to the shock represented by the adoption of the IRB approach. Consistent with *H3*,  $\beta_3$  should be significant and positive (negative) in the average cost (the total amount) of credit model.

#### 5.2 Descriptive statistics

Table 1 shows a set of descriptive statistics on credit relationships by comparing treated and control banks. During the period analyzed, the cost and the amount of credit decreased on average both for the low and high-risk firms included in the sample. However, the cost (amount) of credit decreased at a lower (higher) rate for riskier firms, mainly for loans granted by treated intermediaries. In addition credit relationships between riskier firms and treated banks have also a slightly higher probability of ending.

Table 2 presents the summary statistics by comparing low and high-risk firms. Statistics for the sample at the bank-firm level (panel A) show that less-risky borrowers have greater size, interest coverage ratio, fixed assets and EBITDA as well as lower leverage compared with riskier firms. The strength of bank-firm relationships (i.e. the share of credit) is approximately equal for both groups. Statistics for the sample at the firm level (panel B) confirm that riskier borrowers experienced a lower decrease in the average cost of credit and a larger reduction in the total amount of bank loans. The share of low and high-risk firms exposed to the adoption of the IRB approach by one of their lenders is about equal, as indicated by the mean values of the variable *IRBexp* for the two groups of borrowers.

Finally, Table 3 shows the set of bank characteristics included in the analysis by comparing control and treated intermediaries. Before the shift, with respect to former lenders, treated banks were on

<sup>&</sup>lt;sup>10</sup> HHI is calculated as the sum of squared share of total credit granted by each bank to firm *i* in  $t_{+1}$ .

<sup>&</sup>lt;sup>11</sup> The results are robust to considering this dummy equal to 1 for all firms that in  $t_{.1}$  had a relationship with a bank that has adopted the IRB approach in  $t_0$  and 0 otherwise.

average larger, relied less on retail deposits, had lower loan and capital ratios, and were more profitable. These differences motivate the inclusion of bank-time fixed effects in the main models to absorb all time-varying heterogeneity between the two groups of lenders.

# 5.3 Preliminary evidence: share of high-risk loans

Before presenting the analyses at the bank-firm level, this section shows a preliminary evidence by examining data aggregated at the bank level. This exercise verifies whether treated banks have partly reallocated credit from riskier to safer borrowers after the regulatory regime shift. Therefore the share of credit to riskier borrowers granted by treated intermediaries is compared with that extended by the control group.

Table 4 presents the results of a regression model estimated by adopting  $\Delta$ *ShareHighRisk*, which is the difference between the ratio of loans to high-risk borrowers to total loans in  $t_{+1}$  and  $t_{-1}$ , as the dependent variable. *IRBshift* is the variable of interest. The sample consists of the amount of credit aggregated at the bank level granted by SA intermediaries and those that have adopted the IRB approach in  $t_0$ . The model is a repeated cross-section: it is estimated for each bank-specific regulatory shift date and it includes bank and year fixed effects. This table as well as the following ones report robust standard errors.

The estimates show that treated banks reduced by about 6 percentage points the share of loans to riskier borrowers after the shift, 3.5 percentage points more than control lenders. This finding are consistent with the hypothesis that the adoption of the IRB approach is associated with a reduction in the exposure to risker borrowers.

#### 6. Results

# 6.1 The impact of the regulatory regime shift on the cost and the amount of credit

Table 5 shows the results obtained from the estimation of Eqs. (1) and (3).<sup>12</sup> Taking into account that interest rates declined on average for all borrowers in the analyzed periods (as shown in Table 1), the estimates in column (1) show that the interest rates applied to less-risky firms by treated banks decreased by 18 bps more than those charged by control institutions to the same companies (-75 and -57 bps, respectively).<sup>13</sup> In the same period, treated intermediaries reduced interest rates by 50 bps less for riskier borrowers than for safer ones (-25 bps); whiles SA intermediaries reduced rates by 41

<sup>&</sup>lt;sup>12</sup> The number of observations in the cost of credit models is lower because TAXIA does not contain information on all loans included in CR, as discussed in Section 4. An unreported robustness check shows that the results are robust to adopting a constant sample in all models.

<sup>&</sup>lt;sup>13</sup> Using the estimates in column (1) of Table 5, for low-risk borrowers, the average decrease in rates on credit by control banks is equal to the constant (-0.566), while that on credit by treated banks is (-0.566-0.183).

bps less for the same borrowers (-16 bps).<sup>14</sup> As a result, the additional effect of the shift on riskier firms (9 bps)<sup>15</sup> is sizable: it is about 18 per cent of the lower reduction in interest rates borne by riskier borrowers with respect to safer ones.<sup>16</sup>

As regards the firm control variables, banks reduced rates more for larger firms, those with greater interest coverage ratio, EBITDA and lower leverage. In addition the negative sign of *Share* suggests that interest rates decreased more for firms with a greater share of total credit held by the lender bank, consistent with the literature on relationship banking.

Columns (2), (3) and (4) confirm that treated banks reduced interest rates to high-risk borrowers less than control intermediaries also after including bank control variables, firm-time fixed effects, and bank-time fixed effects, respectively. However, the magnitude of the effect declines after saturating the specification (from 9 to 4 bps; column 4). Overall, these results are consistent with H1, implying that a shift in the regulatory approach adopted by the lender had an asymmetric impact on the cost of credit for high and low-risk firms.

Table 6 reports the results obtained from the estimation of Eqs. (2) and (4). The first column indicates that the credit growth of treated intermediaries was 9 percentage points lower than that of control banks both for low and high-risk borrowers in the considered period, as the interaction between *HighRisk* and *IRBshift* is negative but not significant. In contrast, the estimates in columns (2), (3), and (4) show that the coefficient of the interaction is significant when controlling for supply-side factors. In particular, the results in column (4) indicate that, in the year after the shift, treated banks reduced credit by 1 percentage point more for riskier borrowers than for safer ones when compared with SA intermediaries, consistent with *H1*.

# 6.2 The impact of the regulatory regime shift on the extensive margin of credit

This section focuses on the impact of the IRB adoption on the number of borrowers before and after the shift (the extensive margin of credit, H2), comparing the probability of ending for credit relationships between treated or control lenders and low or high-risk borrowers.

Table 7 shows the results of Eqs. (5) and (6) estimated by adopting *Exit* as the dependent variable, as described in Section 5. The estimates in column (1) show that the probability of ending for relationships between treated lenders and less-risky borrowers in the year after the shift was about 2 percentage points lower than that for relationships with control intermediaries. In contrast, the probability of ending for relationships with treated banks was 4.4 percentage points higher for riskier

 $<sup>^{14}</sup>$  Using the estimates in column (1), for high-risk borrowers, the average decrease in rates on credit by control banks is equal to (0.411-0.566), while that on credit by treated banks is (-0.566+0.411+0.088).

<sup>&</sup>lt;sup>15</sup> Using the abovementioned changes in the interest rates: (-75+25)-(-57+16) = -9.

<sup>&</sup>lt;sup>16</sup> It is calculated as: 9/50.

firms than for safer ones when compared with that for relationships with control intermediaries, consistent with H2. Columns (2), (3), and (4) confirm that the results are robust to including bank control variables as well as firm-time and bank-time fixed effects.

Overall, these findings suggest that the adoption of the IRB approach had a significant impact on the extensive margin of credit, which imply that existing relationships with treated banks are more likely to end if the loan was granted to a riskier borrower (H2).

#### 6.3 The impact of the regulatory regime shift at the firm level

The framework employed in previous sections does not allow taking into account the overall impact at the firm level. Indeed, riskier borrowers may offset the negative effects on the cost and the amount of credit associated with the regulatory shift of their lenders by relying on control banks. The exercise presented in this section allows verifying this hypothesis.

Columns (1) and (2) of Table 8 report the results of Eqs. (7) and (8), respectively. For borrowers more exposed to a treated bank before its shift (i.e. a share of credit granted by a treated bank above the 75<sup>th</sup> percentile of the distribution), the estimates presented in column (1) document that high-risk firms experienced a decrease in the average interest rates that is about 43 bps lower than that faced by safer companies.<sup>17</sup> This is 17 bps higher than that observed for borrowers of SA banks in the same period (26 bps). As indicated in column (2), riskier borrowers of a treated intermediary also experienced a statistically significant greater reduction in the total bank credit growth, but the differential effect is economically negligible (about 0.5 percentage points).

These findings are consistent with hypothesis *H3*, although evidence is weaker for loan amounts. Interestingly, the firm-level effect on loan rates for riskier borrowers (17 bps) is larger than the impact previously shown at the bank-firm level (4 bps). This result may be due to the costs incurred by those borrowers that switched from treated to control banks after the shift. Indeed, since the decrease in total credit is negligible (as shown in column (2) of Table 8), the estimates suggest that riskier borrowers that experienced a reduction in credit growth or a relationship ending by a treated intermediary were able to obtain funds from SA banks; however, new lenders charged higher rates. This additional spread payed by high-risk firms is consistent with the evidence on the significant switching costs for riskier borrowers documented in the literature (Rajan, 1992; Sharpe, 1990).

To verify whether the increase in loan rates is connected with higher switching costs for riskier borrowers, Eq. (5) is estimated by interacting *ShareHHI* with *HighRisk* and *IRBexp*. Indeed, since *ShareHHI* indicates how the credit is concentrated for each firm, this variable can be employed as a

<sup>&</sup>lt;sup>17</sup> Using the estimates in column (1): (0.263-0.074+0.171)-(-0.074) = 0.434.

proxy of the hold-up power of lenders (Ioannidou and Ongena, 2010). Column (3) of Table 8 presents the results of this exercise. The positive sign of the *HighRisk·IRBexp·ShareHHI* coefficient suggests that the decrease in the average interest rates was lower mainly for riskier borrowers of treated banks that had a highly concentrated credit (i.e. high concentration index). This evidence implies that the costs for firms involved in switching lenders may strengthen the negative impact of the IRB adoption on interest rates.

#### 7. Additional tests

The following sections present additional estimates to strengthen the validity of the main findings. In the first set of tests, the variable *HighRisk* is replaced with different borrower-level variables. The second robustness check presents a falsification test. For space considerations, the following tests focus on the cost of credit model (Eq. 3), but the results are also robust to adopting Eq. (4).

Moreover, unreported robustness checks show that the results are robust to examining the yearly granted credit, considering the overall credit extended by each bank (the sum of credit lines, term loans, and loans backed by accounts receivable), and focusing only on the shifts occurred after the global financial crisis.

#### 7.1. Borrowers' characteristics

The main models presented in previous sections are based on an accounting credit score to identify riskier borrowers. This section focuses on different borrowers' characteristics to explore the categories of firms that are more affected by a regulatory regime shift.

First, the level of debt may represent a proxy of the borrower's riskiness: highly leveraged firms are generally more financially constrained and may not be able to repay their obligations. Column (1) of Table 9 shows the results of Eq. (3) estimated by replacing *HighRisk* with *HighLev*, which is a dummy variable equal to 1 if the firm's leverage is greater than the median in  $t_0$  and 0 otherwise. The results document that highly leveraged firms experienced a lower decrease in rates on credit granted by treated banks.

The second test focuses on the borrower size. Smaller firms are considered as riskier than larger ones because they are on average financially weaker and less transparent (Petersen and Rajan, 1994; Rajan and Zingales, 1995). Column (2) of Table 9 shows the results of Eq. (3) estimated by replacing *HighRisk* with *Smaller*, which is a dummy variable equal to 1 if the firm' total assets are lower than the median in  $t_0$  and 0 otherwise. The estimates indicate that treated banks decreased interest rates less for smaller borrowers.

Overall, these tests suggest that the main findings are robust to adopting different measures to identify riskier borrowers.

#### 7.2. Falsification test

This section presents a falsification test to address potential concerns about structural differences in lending policies among treated and control banks that might exist before the regulatory regime shift.

To employ this test, Eq. (3) is estimated by considering the year before the real shift date as  $t_0$ . Therefore, this test is equivalent to assume that the shift date has occurred in  $t_{-1}$  for each treated bank. Table 10 shows the results of this falsification exercise. The interaction coefficient between *HighRisk* and *IRBshift* is not significant, suggesting that the change in interest rates for high-risk borrowers with respect to less-risky ones is not significantly different between treated and control banks before the shift date.

#### 8. Conclusions

This paper analyses whether the adoption of internal rating models affects banks' lending policies for low and high-risk borrowers. The main hypothesis is that the relatively disadvantageous treatment that loans for riskier firms receives under the IRB approach may lead banks adopting this method to raise interest rates and reduce credit to high-risk borrowers compared with low-risk ones.

The empirical evidence indicates that, in the year after the regulatory regime shift, banks adopting the IRB approach reduce interest rates (credit) to riskier borrowers compared with less-risky ones less (more) than SA institutions. Second, an existing relationship with a bank adopting the IRB method is more likely to end if the loan is granted to a high-risk borrower. Third, after the regulatory regime shift, the results obtained at the firm-level suggest that riskier firms were able to compensate partly the reduction in bank credit by obtaining funds from SA banks, but they could not offset the negative effect on the average cost of credit because of the significant costs associated with switching lenders.

The results are robust to adopting the firm's level of debt or size to identify riskier borrowers and employing a falsification test.

These findings are associated with several policy implications. The results suggest that an increase in the number of intermediaries adopting the IRB method may incentivize a safer allocation of credit. Indeed, as shown by the bank-level analysis in Section 5.3, a regulatory shift is associated with a greater reduction in the share of loans to riskier borrowers. A side effect of this process is that the access to credit for riskier firms may be hampered by an increase in the average cost of credit, which may reach an unsustainable level for already financially constrained borrowers.

Overall, it should be noted that the presented analysis does not provide a comprehensive assessment of the effects of risk-based capital requirements for the whole financial system, while it focuses only on the impact of the IRB adoption on bank lending policies for high and low-risk firms. Therefore, further research on this topic is required to a more accurate assessment of the overall costs and benefits of this regulatory framework.

# Appendix

# Table A.1

Variables description.

Variable	Description	Source
Dependent Variabl	e	
$\Delta Rate$	Difference between the annualized rate on granted credit lines in $t_{+1}$ and $t_{-1}$ by bank j to firm i.	CR
$\Delta Credit$	Difference between the log amount of granted credit lines in $t_{+1}$ and $t_{-1}$ by bank <i>j</i> to firm <i>i</i> .	CR
Exit	Dummy variable equal to 1 for loans that existed in <i>t</i> -1 and ceased to exist after the shift.	CR
$\Delta AvgRate$	Difference between the average annualized rate on granted credit lines in $t_{+1}$ and $t_{-1}$ to firm <i>i</i> .	CR
$\Delta TotCredit$	Difference between the log amount of total granted credit lines in $t_{+1}$ and $t_{-1}$ to firm <i>i</i> .	CR
$\Delta Share High Risk$	Difference between the ratio of loans to high-risk borrowers to total loans in $t_{+1}$ and $t_{-1}$ .	CR
Key Explanatory V	Variables	
HighRisk	Dummy variable equal to 1 if the firm's credit score in $t_0$ is greater than 5 and 0 otherwise.	Cerved
IRBshift	Dummy variable equal to 1 if the bank $j$ has adopted the IRB approach in $t_0$ and 0 otherwise.	Supervisory and Pillar 3 information
IRBexp	Dummy variable equal to 1 for firms that in <i>t</i> -1 had a share of credit granted by a treated bank falling in the top quartile of the distribution and 0 otherwise.	Supervisory and Pillar 3 information
HighLev	Dummy variable equal to 1 if the firm's leverage is greater than the median in $t_0$ and 0 otherwise.	Cerved
Smaller	Dummy variable equal to 1 if the firm's total assets are lower than the median in $t_0$ and 0 otherwise.	Cerved
Control Variables		
Share	The share of total credit (revolving credit lines, term loans, loans backed by accounts receivables) granted by bank <i>i</i> to firm <i>i</i> in $t_{\pm i}$ .	CR
ShareHHI	The Herfindahl-Hirschman index (HHI) of share of total credit granted to firm <i>i</i> in $t_{+1}$ .	CR
BankSize	Logarithm of the total assets of bank $j$ in $t_{-1}$ .	Supervisory
Loans	Ratio of total loans to total assets of bank $j$ in $t$ - $1$ .	reports Supervisory reports
Deposits	Ratio of deposits to total liabilities of bank <i>j</i> in <i>t</i> -1.	Supervisory
CapRatio	Ratio of total equity to total assets of bank $j$ in $t_{-1}$ .	Supervisory
ROE	Return on equity (pre-tax income over total equity) of bank $j$ in $t_{-1}$ .	Supervisory reports
Size	Logarithm of the total assets of firm $i$ in $t_0$ .	Cerved
IntCovRatio	Ratio of EBITDA to interest expense of firm $i$ in $t_0$ .	Cerved
Leverage	Ratio of total debt to the book value of assets of firm $i$ in $t_0$ .	Cerved
Fixed Assets	Ratio of fixed assets to total assets of firm $i$ in $t_0$ .	Cerved
EBITDA	Ratio of EBITDA to total assets of firm $i$ in $t_0$ .	Cerved
Industry	Indicator variables for the industry of firm <i>i</i> based on 2-digit ATECO.	Cerved
Area	Indicator variables for area of the country: north-west, north-east, center, south.	Cerved

# References

- Altman, E.I., Marco, G., Varetto, F., 1994. Corporate distress diagnosis: Comparisons using linear discriminant analysis and neural networks (the Italian experience). Journal of Banking and Finance 18, 505–529.
- BCBS, 2017. High-level summary of Basel III reforms.
- Behn, M., Haselmann, R., Vig, V., 2016a. The limits of model-based regulation, ECB Working Paper Series.
- Behn, M., Haselmann, R., Wachtel, P., 2016b. Procyclical Capital Regulation and Lending. Journal of Finance 71, 919–956.
- Boot, A.W.A., 2000. Relationship Banking: What Do We Know? Journal of Financial Intermediation 9, 7–25.
- Bruno, B., Nocera, G., Resti, A., 2017. Are risk-based capital requirements detrimental to corporate lending? Evidence from europe, CEPR working paper.
- Cannata, F., Casellina, S., Guidi, G., 2012. Inside the Labyrinth of Basel Risk-Weighted Assets: How Not to Get Lost, Questioni di Economia e Finanza (Occasional Papers), Bank of Italy.
- Cucinelli, D., Di Battista, M.L., Marchese, M., Nieri, L., 2018. Credit risk in European banks: The bright side of the internal ratings based approach. Journal of Banking and Finance 93, 213–229.
- EBA, 2013. Report on the pro-cyclicality of capital requirements under the Internal Ratings Based Approach.
- Elsas, R., 2005. Empirical determinants of relationship lending. Journal of Financial Intermediation 14, 32–57.
- Gallo, R., 2020. The impact of the IRB approach on the relationship between the cost of credit for public companies and financial market conditions, Temi di Discussione (Working Papers), Bank of Italy.
- Ioannidou, V., Ongena, S., 2010. "Time for a change": Loan conditions and bank behavior when firms switch banks. Journal of Finance 65, 1847–1877.
- Jiménez, G., Lopez, J.A., Saurina, J., 2009. Empirical analysis of corporate credit lines. Review of Financial Studies 22, 5069–5098.
- Jiménez, G., Ongena, S., Peydró, J.-L., Saurina, J., 2014. Hazardous Times for Monetary Policy:

What Do Twenty-Three Million Bank Loans Say About The Effects of Monetary Policy on Credit Risk-Taking? Econometrica 82, 463–505.

- Kashyap, A., Stein, J., 2004. Cyclical implications of the Basel II capital standards. Federal Reserve Bank of Chicago 28, 18–31.
- Khwaja, A.I., Mian, A., 2008. Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market. American Economic Review 98, 1413–1442.
- Le Leslé, V., Avramova, S., 2012. Revisiting Risk-Weighted Assets "Why Do RWAs Differ Across Countries and What Can Be Done About It?," IMF Working Paper.
- Mariathasan, M., Merrouche, O., 2014. The manipulation of basel risk-weights. Journal of Financial Intermediation 23, 300–321.
- Pérez Montes, C., Trucharte Artigas, C., Cristófoli, M.E., Lavín San Segundo, N., 2018. The impact of the IRB approach on the risk weights of European banks. Journal of Financial Stability 39, 147–166.
- Petersen, M.A., Rajan, R.G., 1994. The Benefits of Lending Relationships: Evidence from Small Business Data. The Journal of Finance 49, 3–37.
- Plosser, M.C., Santos, J.A.C., 2014. Banks' incentives and the quality of internal risk models, Federal Reserve Bank of New York Staff Reports.
- Rajan, R.G., 1992. Insiders and Outsiders: The Choice between Informed and Arm's Length Debt. The Journal of Finance 47, 1367–1400.
- Rajan, R.G., Zingales, L., 1995. What Do We Know about Capital Structure? Some Evidence from International Data. The Journal of Finance 50, 1421–1460.
- Repullo, R., Suarez, J., 2004. Loan pricing under Basel capital requirements. Journal of Financial Intermediation 13, 496–521.
- Repullo, R., Suarez, J., 2013. The procyclical effects of bank capital regulation. Review of Financial Studies 26, 452–490.
- Sette, E., Gobbi, G., 2015. Relationship lending during a financial crisis. Journal of the European Economic Association 13, 453–481.

Sharpe, S., 1990. Asymmetric Information, Bank Lending, and Implicit Contracts: A Stylized Model of Customer Relationships. The Journal of Finance 45, 1069–1087.

# **Figures and tables**

#### Figure 1

Number of IRB banks by year (2008-17).



### Figure 2

Share of the overall granted credit to non-financial firms by regulatory approach in the examined period.



#### Table 1

Summary statistics on credit relationships comparing treated and control banks

		(	<b>Control Banks</b>			reated Bar	ıks
		Mean	Median	Std. Dev.	Mean	Median	Std. Dev.
ΔRate	LR	-0.72	-0.33	2.37	-1.44	-1.30	2.44
	HR	-0.18	0.01	2.46	-0.81	-0.54	2.40
∆Credit	LR	-0.37	0.00	58.51	-6.39	0.00	66.60
( <i>p.p.</i> )	HR	-2.77	0.00	60.94	-9.07	0.00	68.44
Errit	LR	0.15	0.00	0.36	0.13	0.00	0.34
EXIL	HR	0.25	0.00	0.43	0.27	0.00	0.44

Summary statistics comparing low and high-risk firms.

Variable	LR Firms				HR Firms	6
-	Mean	Median	Std. Dev.	Mean	Median	Std. Dev.
A: Bank-Firm Level						
∆Rate	-1.00	-0.67	2.44	-0.44	-0.15	2.46
$\Delta$ Credit ( <i>p.p.</i> )	-2.41	0.00	61.44	-5.02	0.00	63.80
IRBshift	0.34	0.00	0.47	0.36	0.00	0.48
Share	0.21	0.16	0.17	0.21	0.16	0.18
Size	9.00	8.90	1.55	8.48	8.36	1.41
IntCovRatio	10.43	5.65	11.73	2.01	1.82	3.16
Leverage	0.69	0.72	0.22	0.87	0.89	0.18
Fixed Assets	0.24	0.18	0.22	0.20	0.12	0.21
EBITDA	0.10	0.08	0.12	0.05	0.05	0.11
B: Firm level						
∆AvgRate	-0.75	-0.51	2.05	-0.29	-0.10	2.07
$\Delta$ TotCredit ( <i>p.p.</i> )	-2.01	0.00	44.06	-8.12	0.00	50.03
IRBexp	0.26	0.00	0.44	0.24	0.00	0.43
ShareHHI	0.61	0.53	0.30	0.63	0.55	0.29

#### Table 3

Summary statistics comparing treated and control banks.

Variable	Treated Banks			(	Control Bar	iks
	Mean	Median	Std. Dev.	Mean	Median	Std. Dev.
BankSize (€bn)	504.11	538.99	373.58	225.93	90.66	299.46
Loans (%)	73.09	71.26	5.74	73.98	74.20	6.57
Deposits (%)	34.43	38.51	10.53	44.70	44.27	12.64
CapRatio (%)	7.31	6.17	1.78	8.00	8.24	2.11
ROE (%)	3.85	6.73	13.19	-1.30	4.85	17.59

#### Table 4

The share of credit granted to high-risk firms.

	(1)
	$\Delta Share High Risk$
IRBshift	-0.035***
	(0.010)
Constant	-0.022***
	(0.000)
Bank FE	Yes
Year FE	Yes
Observations	1,956
Adj R-squared	0.216

The table shows the results of a regression model estimated by examining the amount of credit aggregated at the bank level granted by SA intermediaries and those that have adopted the IRB approach in  $t_0$ . The dependent variable is  $\Delta$ *ShareHighRisk*, the difference between the ratio of loans to high-risk borrowers to total loans in  $t_{+1}$  and  $t_{-1}$ . Robust *p*-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

The impact of the IRB adoption on the cost of credit.

	(1)	(2)	(3)	(4)
	$\Delta Rate$	$\Delta Rate$	$\Delta Rate$	∆Rate
HighRisk	0.411***	0.444***	0.396***	-
	(0.000)	(0.000)	(0.000)	
IRBshift	-0.183***	-0.377***	-	-
W 15 1 155 1 4	(0.000)	(0.000)		0.000
HighRisk-IRBshift	0.088***	0.133***	0.114***	0.039**
Ch and	(0.000)	(0.000)	(0.000)	(0.031)
Snare	-0.469****	-0.453****	-0.514****	-
Size	-0.026***	-0.032***	-0.032***	_
Size	-0.020	(0.000)	(0,000)	
IntCovRat	-0.016***	-0.012***	-0.015***	-
	(0.000)	(0.000)	(0.000)	
Leverage	0.200***	0.004	0.208***	-
	(0.000)	(0.325)	(0.000)	
FixedAsset	-0.016	-0.023	-0.011	-
	(0.484)	(0.282)	(0.648)	
EBITDA	-0.848***	-0.987***	-0.871***	-
D 1.0'	(0.000)	(0.000)	(0.000)	
BankSize	-	0.141***	-	-
Loons		(0.000)		
Loans	-	(0.001)	-	-
Deposits	_	0 361***	-	_
Deposito		(0.000)		
CapRatio	-	1.452***	-	-
		(0.000)		
ROE	-	-0.512***	-	-
		(0.000)		
Constant	-0.566***	-1.380***	-0.577***	-0.619***
	(0.000)	(0.000)	(0.000)	(0.000)
Year, Industry, Area FE	Yes	Yes	Yes	No
Firm-Time FE	No	No	No	Yes
Bank-Time FE	No	No	Yes	Yes
Observations	341,370	341,370	341,370	341,370
Adj R-squared	0.164	0.168	0.187	0.293

Columns (1), (2) and (3) show the results obtained from the estimation of Eq. (1) by adding firm controls, bank controls, and banktime FE, respectively. Column 4 presents the results of Eq. (3). The dependent variable is  $\Delta Rate$ , the difference between the annualized rate on granted credit lines in  $t_{+1}$  and  $t_{-1}$  by bank j to firm i. Robust p-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

The impact of the IRB adoption on the amount of credit.

	(1)	(2)	(3)	(4)
	$\Delta Credit$	∆Credit	∆Credit	∆Credit
HighRisk	-0.026***	-0.023***	-0.026***	-
	(0.000)	(0.000)	(0.000)	
IRBshift	-0.087***	-0.042***	-	-
	(0.000)	(0.000)		
HighRisk·IRBshift	-0.004	-0.007*	-0.007*	-0.010***
	(0.265)	(0.057)	(0.091)	(0.010)
Constant	-0.048***	-0.055**	-0.164***	-0.035***
	(0.000)	(0.010)	(0.000)	(0.000)
Firm Controls	Yes	Yes	Yes	No
Bank Controls	No	Yes	No	No
Firm-Time FE	No	No	No	Yes
Bank-Time FE	No	No	Yes	Yes
Observations	599,418	599,418	599,418	599,418
Adj R-squared	0.011	0.014	0.033	0.055

Columns (1), (2) and (3) show the results obtained from the estimation of Eq. (2) by adding firm controls, bank controls, and banktime FE, respectively. Column 4 presents the results of Eq. (4). The dependent variable is  $\Delta Credit$ , difference between the log amount of total granted credit lines in  $t_{+1}$  and  $t_{-1}$  to firm *i*. Robust *p*-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

#### Table 7

The impact of the IRB adoption on the "extensive margin" of credit.

	(1)	(2)	(3)	(4)
	Exit	Exit	Exit	Exit
HighRisk	0.054***	0.051***	0.054***	-
	(0.000)	(0.000)	(0.000)	
IRBshift	-0.021***	-0.018***	-	-
	(0.000)	(0.000)		
HighRisk·IRBshift	0.044***	$0.044^{***}$	0.036***	0.045***
	(0.000)	(0.000)	(0.000)	(0.000)
Constant	0.523***	1.167***	0.514***	0.198***
	(0.000)	(0.000)	(0.000)	(0.000)
Firm Controls	Yes	Yes	Yes	No
Bank Controls	No	Yes	No	No
Firm-Time FE	No	No	No	Yes
Bank-Time FE	No	No	Yes	Yes
Observations	459,345	459,345	459,345	459,345
Adj R-squared	0.053	0.064	0.152	0.345

Columns (1), (2) and (3) show the results obtained from the estimation of Eq. (5) by adding firm controls, bank controls, and banktime FE, respectively. Column 4 presents the results of Eq. (6). The sample includes firms with at least one loan granted by a treated and a control bank in  $t_{-1}$ . Robust *p*-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

The impact of the IRB adoption on the cost and the amount of credit at the firm level.

	(1)	(2)	(3)
	$\Delta AvgRate$	$\Delta TotCredit$	$\Delta AvgRate$
HighRisk	0.263***	-0.061***	0.358***
	(0.000)	(0.000)	(0.000)
IRBexp	-0.074***	-0.024***	-0.064***
	(0.000)	(0.000)	(0.001)
HighRisk·IRBexp	0.171***	-0.005*	0.046*
	(0.000)	(0.051)	(0.063)
HighRisk·ShareHHI	-	-	-0.167***
			(0.000)
IRBexp·ShareHHI	-	-	-0.009
			(0.804)
HighRisk·IRBexp·ShareHHI	-	-	0.224***
			(0.000)
ShareHHI	0.207***	-0.270***	0.288***
	(0.000)	(0.000)	(0.000)
Constant	-0.726***	0.525***	-0.780***
Constant	(0,000)	(0,000)	(0,000)
	(0.000)	(0.000)	(0.000)
Firm Controls	Yes	Yes	Yes
Observations	640,320	1,048,032	640,320
Adj R-squared	0.145	0.046	0.145

Column (1) presents the results of Eq. (7). The dependent variable of this model is  $\Delta AvgRate$ , the difference between the average annualized rate on granted credit lines in  $t_{+1}$  and  $t_{-1}$  to firm *i*. Column (2) presents the results of Eq. (8). The dependent variable of this model is  $\Delta TotCredit$ , the difference between the log amount of total granted credit lines in  $t_{+1}$  and  $t_{-1}$  to firm *i*. In Column (3), Eq. (7) is estimated by interacting *ShareHHI* with *HighRisk* and *IRBexp*. Robust *p*-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

#### Table 9

The impact of the IRB adoption on the cost of credit by exploring borrowers' characteristics.

	(1)	(2)
	Leverage	Size
HighLev·IRBshift	0.048***	-
	(0.006)	
Smaller·IRBshift	-	0.045**
		(0.011)
Constant	-0.618***	-0.618***
	(0.000)	(0.000)
Firm-Time FE	Yes	Yes
Bank-Time FE	Yes	Yes
Observations	341,370	341,370
Adj R-squared	0.293	0.293

Columns (1) and (2) present the results of Eq. (3) estimated by replacing *HighRisk* with *HighLev* and *Smaller*, respectively. The dependent variable is  $\Delta Rate$ , the difference between the annualized rate on granted credit lines in  $t_{+1}$  and  $t_{-1}$  by bank *j* to firm *i*. Robust *p*-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

Falsification test.

	(1)
	$\Delta Rate$
HighRisk·IRBshift	0.020
	(0.164)
Constant	0.648***
	(0.000)
Firm-Time FE	Yes
Bank-Time FE	Yes
Observations	267,619
Adj R-squared	0.381

The table shows the results of the falsification test. The dependent variable is  $\Delta Rate$ , the difference between the annualized rate on granted credit lines in  $t_{+1}$  and  $t_{-1}$  by bank *j* to firm *i*. Robust *p*-values in parentheses. \*\*\*, \*\*, and \* denote significance at the 1%, 5%, and 10% levels, respectively.

- N. 1322 Judicial efficiency and bank credit to firms, by Giacomo Rodano (March 2021).
- N. 1323 Unconventional monetary policies and expectations on economic variables, by Alessio Anzuini and Luca Rossi (March 2021).
- N. 1324 *Modeling and forecasting macroeconomic dowside risk*, by Davide Delle Monache, Andrea De Polis and Ivan Petrella (March 2021).
- N. 1325 Foreclosures and house prices, by Michele Loberto (March 2021).
- N. 1326 *inancial structure and bank relationships of Italian multinational firms*, by Raffaello Bronzini, Alessio D'Ignazio and Davide Revelli (March 2021).
- N.1327 Foreign investors and target firms' financial structure: cavalry or lucusts?, by Lorenzo Bencivelli and Beniamino Pisicoli (April 2021).
- N. 1328 Board composition and performance of state-owned enterprises: quasi experimental evidence, by Audinga Baltrunaite, Mario Cannella, Sauro Mocetti and Giacomo Roma (April 2021).
- N. 1329 Can internet banking affect households' participation in financial markets and financial awareness?, by Valentina Michelangeli and Eliana Viviano (April 2021).
- N. 1330 (In)Efficient separations, firing costs and temporary contracts, by Andrea Gerali, Elisa Guglielminetti and Danilo Liberati (April 2021).
- N. 1331 *The catalytic role of IMF programs*, by Claudia Maurini and Alessandro Schiavone (April 2021).
- N.1332 Dating the euro area business cycle: an evaluation, by Claudia Pacella (April 2021).
- N. 1333 Population aging, relative prices and capital flows across the globe, by Andrea Papetti (April 2021).
- N.1334 What drives investors to chase returns?, by Jonathan Huntley, Valentina Michelangeli and Felix Reichling (April 2021).
- N. 1335 Managerial talent and managerial practices: are they complements?, by Audinga Baltrunaite, Giulia Bovini and Sauro Mocetti (April 2021).
- N.1336 Volatility bursts: a discrete-time option model with multiple volatility components, by Francesca Lilla (June 2021).
- N.1337 A liquidity risk early warning indicator for Italian banks: a machine learning approach, by Maria Ludovica Drudi and Stefano Nobili (June 2021).
- N.1338 Macroprudential policy analysis via an agent based model of the real estate sector, by Gennaro Catapano, Francesco Franceschi, Michele Loberto and Valentina Michelangeli (June 2021).
- N.1339 Optimal robust monetary policy with parameters and output gap uncertainty, by Adriana Grasso and Guido Traficante (June 2021).
- N.1340 Information or persuasion in the mortgage market: the role of brand names, by Agnese Carella and Valentina Michelangeli (June 2021).
- N.1341 *The grocery trolley race in times of Covid-19: evidence from Italy*, by Emanuela Ciapanna and Gabriele Rovigatti (June 2021).
- N. 1342 When the panic broke out: COVID-19 and investment funds' portfolio rebalancing around the world, by Massimiliano Affinito and Raffaele Santioni (July 2021).
- N.1343 *Firm-bank linkages and optimal policies in a lockdown*, by Anatoli Segura and Alonso Villacorta (July 2021).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform,* European Journal of Political Economy, v. 59, pp. 423-444, WP 1135 (September 2017).
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, *Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis,* Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, **WP 1086 (October 2016).**
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area*, IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, **WP 1186 (July 2018).**
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme,* Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, WP 1185 (July 2018).
- BUSETTI F. and M. CAIVANO, *Low frequency drivers of the real interest rate: empirical evidence for advanced economies*, International Finance, v. 22, 2, pp. 171-185, **WP 1132 (September 2017).**
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, *Forecasting with instabilities: an application to DSGE models with financial frictions*, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, *Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears,* Economic Modelling, v. 77, pp. 266-275, WP 1076 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, WP 836 (November 2011).
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side and demand-side factors,* Social Indicators Research, v. 143, 3, pp. 1185–1217, WP 989 (November 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme,* IMF Economic Review, v. 67, 2, pp. 315-348, WP 1036 (October 2015).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic,* Journal of Macroeconomics, v. 62, article 103019, WP 1140 (September 2017).
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, WP 1167 (March 2018).
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*, 1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, WP 1127 (July 2017).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, v. 46, article 101632, WP 1105 (March 2017).
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MERCATANTI A., T. MAKINEN and A. SILVESTRINI, *The role of financial factors for european corporate investment,* Journal of International Money and Finance, v. 96, pp. 246-258, WP 1148 (October 2017).
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, Short-term forecasts of economic activity: are fortnightly factors useful?, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, v. 46, 4, WP 984 (November 2014).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, WP 1047 (January 2016).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times*, R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, *Public debt expansions and the dynamics of the household borrowing constraint,* Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities,* Journal of Business & Economic Statistics, , v. 38, 4, pp. 826-838, WP 974 (October 2014).
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy*, Journal of International Economics, v. 126, article 103328, **WP 1032** (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, *Services trade and credit frictions: evidence with matched bank firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, v. 35, 3, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome*, Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis,* Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, WP 1179 (June 2018).
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises,* Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, **WP 1299 (October 2020).**
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case,* Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).

- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, WP 1059 (March 2016).
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, WP 1072 (July 2016).

#### 2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019).**
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- BARONE G., F. DAVID, G. DE BLASIO and S.MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (Oct 2019).
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, WP 1187 (September 2018).
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, *Domestic and global determinants of inflation: evidence from expectile regression*, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1324 (March 2021).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit*, Journal of Regional Science, v. 61, 3, pp. 597-622, WP 1222 (June 2019).
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, *Secular stagnation, R&D, public investment and monetary policy: a global-model perspective,* Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).
- DEL PRETE S. and M. L. STEFANI, Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, WP 1014 (June 2015).
- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area*, Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds*, The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**

- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, WP 1095 (January 2017).
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

#### FORTHCOMING

- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence* from the Italian Mezzogiorno, Papers in Regional Science, **WP 1253 (December 2019).**
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, WP 1197 (November 2018).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, WP 1295 (September 2020).
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, **WP 1133 (September 2017).**
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, WP 1269 (March 2020).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, **WP 1288 (July 2020).**
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, *Monetary Policy strategies in the new normal: a model*based analysis for the Euro Area, Journal of Macroeconomics, v. 70, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, WP 1224 (June 2019).
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, **WP 1119 (June 2017).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, **WP 1101 (February 2017).**
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, **WP 1252 (December 2019).**
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers*, Journal of Money, Credit and Banking, **WP 1277 (June 2020).**
- LOBERTO M, *Foreclosures and house prices*, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, **WP 1171 (April 2018).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, WP 1182 (July 2018).
- PAPETTI A., Demographics and the natural real interest rate: historical and projected paths for the Euro Area, Journal of Economic Dynamics and Control, WP 1306 (November 2020).
- PERICOLI M. and M. TABOGA, Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models, Journal of Financial Econometrics, WP 1189 (September 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, WP 1142 (October 2017).
- SCHIVARDI F., E. SETTE and G. TABELLINI, Credit misallocation during the European financial crisis, Economic Journal, WP 1139 (September 2017).
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, WP 1243 (November 2019).